Author's response to reviews

Title: Maternal near-miss and death and their association with caesarean section complications: a cross-sectional study at a university hospital and a regional hospital in Tanzania

Authors:

Helena Litorp (helena.litorp@kbh.uu.se)
Hussein L Kidanto (hkidanto@yahoo.co.uk)
Mattias Rööst (mattias.roost@kbh.uu.se)
Muzdalifat Abeid (muzsalim@yahoo.com)
Lennarth Nyström (lennarth.nystrom@epiph.umu.se)
Birgitta Essén (birgitta.essen@kbh.uu.se)

Version: 8 Date: 12 June 2014

Author's response to reviews: see over
12 June 2014

Dear Dr. van den Akker:

We are glad that you have found our manuscript of interest to readers of BMC Pregnancy and Childbirth. Please find the requested list of abbreviations in the manuscript, preceding “Competing interests” and “Authors' contributions”.

Sincerely,

Helena Litorp, M.D
Department of Women’s and Children’s Health
Uppsala University
Uppsala University Hospital
751 85 Uppsala
Sweden

E-mail: helena.litorp@kbh.uu.se
Tel: +46 70 254 27 94